Author's response to reviews

Title: CONNECT for Quality: Protocol of a Randomized Controlled Trial to Improve Fall Prevention in Nursing Homes

Authors:

Ruth A Anderson (ruth.anderson@duke.edu)
Kirsten Corazzini (coraz001@mc.duke.edu)
Kristie Porter (Kirstie.porter@duke.edu)
Katherine Daily (katherine.daily@duke.edu)
Reuben R McDaniel Jr. (Reuben.McDaniel@mccombs.utexas.edu)
Cathleen Colon-Emeric (Colon001@mc.duke.edu)

Version: 2 Date: 19 September 2011

Author's response to reviews: see over
Mr Paul M Wilson  
The Implementation Science Editorial Team  
BioMed Central Ltd  
Floor 6, 236 Gray's Inn Road  
London WC1X 8HL  
United Kingdom

September 3, 2011

Dear Mr. Wilson,

We have revised the protocol entitled, CONNECT for Quality: Protocol of a Randomized Controlled Trial to Improve Fall Prevention in Nursing Home and added the CONSORT checklist, paying particular attention to items 3-13 as you requested. Please note however, that item 13 is not relevant because we have not yet begun enrolling participants. We added the page numbers to the checklist. The changes to the manuscript were minimal—adding one sentence about how we assessed the blinding of the data collectors of the outcome variables.

Again, thank you for the opportunity to submit our work to Implementation Science. We look forward to your review.

Respectfully,

Ruth A. Anderson, RN, PhD, FAAN  
Virginia Stone Professor of Nursing  
Research Development Coordinator, Office of Research Affairs  
Senior Fellow, Center for the Study of Aging and Human Development  
Duke University School of Nursing  
Durham, NC 27710  
Office: 919-668-4599  
ruth.anderson@duke.edu